The effect of board diversity and tenure on environmental performance. Evidence from family and non-family firms

Giovanna Gavana Department of Economics, University of Insubria, Varese, Italy, and Pietro Gottardo and Anna Maria Moisello

Department of Economics and Management, University of Pavia, Pavia, Italy

Abstract

Purpose – The aim of this paper is to examine the effect of structural and demographic board diversity as well as board tenure on family firms' environmental performance, by analyzing the differences between family and non-family businesses and within family firms.

Design/methodology/approach - Tobit regressions are applied to investigate the effect of independent directors, CEO non-duality, board gender diversity and board tenure on environmental performance. The study also controls for other board and firm characteristics, as well as for time, industry and country-fixed effects. In doing so, the authors rely on a sample of non-financial listed firms from France, Germany, Italy, Spain and Portugal over the period 2014–2021.

Findings - The authors find that women on the board positively influence environmental performance and this effect is significant only in family firms, although board tenure negatively moderates the relationship. Board independence significantly affects environmental performance only in non-family firms. A strong presence of family directors has a negative effect on family firms' environmental performance, especially when directors' turnover is low.

Originality/value - This paper examines the unexplored relationship between structural board diversity and environmental performance in family companies. This study provides empirical evidence on the association between gender diversity and family firms' environmental performance focusing for the first time on a European setting. Moreover, this study provides evidence of a different effect of board tenure in family and non-family businesses

Keywords Environmental performance, Family firms, Board independence, Board gender diversity, CEO duality, Board tenure

Paper type Research paper

1. Introduction

Preservation of the environment has become a major concern over recent decades and companies face increasing societal pressure to reduce their environmental footprint. Legislative interventions, worldwide, aim at enhancing firms' environmental sustainability and directing investments towards economic activities that substantially contribute to the mitigation of climate change and, more generally, to the protection of the natural environment, Literature has also highlighted the role of environmental operational practices in enhancing a firm's competitive advantage (Rousseau et al., 2019). Therefore, environmental engagement challenges corporate governance as boards of directors play a fundamental role in establishing strategic

© Giovanna Gavana, Pietro Gottardo and Anna Maria Moisello. Published by Emerald Publishing Limited. This article is published under the Creative Commons Attribution (CC BY 4.0) licence. Anyone may reproduce, distribute, translate and create derivative works of this article (for both commercial and non-commercial purposes), subject to full attribution to the original publication and authors. The full terms of this licence may be seen at http://creativecommons.org/licences/by/4.0/legalcode

Journal of Family Business Management Emerald Publishing Limited 2043-6238 DOI 10.1108/JFBM-06-2023-0088

Received 23 June 2023 Revised 7 September 2023 Accepted 7 September 2023

Effect of board diversity and

tenure

objectives and are ultimately responsible for a firm's financial as well as social and environmental performance (Veltri *et al.*, 2021). Environmental performance constitutes a relevant dimension of operations performance (de Burgos Jiménez and Céspedes Lorente, 2001) as it measures the impact of a company's environmental strategies, in terms of reduction of natural resource consumption, emissions and environmental costs for customers. Boards are diverse in terms of their structural characteristics, such as size, board independence, CEO duality and tenure and their members' demographic attributes such as gender, age, education and nationality. Literature refers to "diversity of boards" when dealing with structural board diversity features and to "diversity in boards" when considering director demographic diversity traits (Veltri *et al.*, 2021). Therefore, research has pointed out the importance of analyzing the effect of specific board characteristics on environmental performance (Hussain *et al.*, 2018).

Previous studies have provided conflicting results regarding the relationship between structural board diversity and environmental performance (De Villiers et al., 2011: Post et al., 2011, 2015; Walls et al., 2012; Arena et al., 2015; Shaukat et al., 2016; Hussain et al., 2018; García Martín and Herrero, 2020; Orazalin and Mahmood, 2021). This stream of research has not taken into account the effect of a firm ownership type addressing the possible differences between family and non-family companies, as well as within family businesses. Very few studies have analyzed the effect of board tenure on environmental performance (De Villiers *et al.*, 2011; Paolone et al., 2023). In particular, no study has verified how tenure may moderate the relationship between board diversity and environmental performance in family firms although there is evidence of a different effect of tenure on family and non-family firms' risk aversion (Huybrechts et al., 2013) as well as financial performance (Tao-Schuchardt and Kammerlander, 2023). Research has also addressed the association between demographic board diversity and environmental performance, mostly focusing on gender diversity, with mixed findings (see, among others: Post et al., 2011; Li et al., 2017; Lu and Herremans, 2019; Francoeur et al., 2019; García Martín and Herrero, 2020; Orazalin and Mahmood, 2021; Islam et al., 2022). A very limited number of studies, focusing on U.K. and U.S. companies, have considered the relationship between demographic board diversity and environmental performance, controlling for a firm's family nature (Cordeiro et al., 2020; Nadeem et al., 2020). Research on family firms' environmental performance has mainly focused on differences in family and non-family companies (Berrone et al., 2010), family involvement in management and level of ownership (Samara et al., 2018; Graafland, 2020; Miroshnychenko et al., 2022). Other studies have investigated peculiar sources of heterogeneity in family firms' environmental performance such as training and development practices (Dal Maso *et al.*, 2020), relations with the social community (Dekker and Hasso, 2016). long-term orientation (Dou et al., 2019) and stakeholder pressure (Neubaum et al., 2012). At present, no study has focused on the effect of board structural diversity on family firms' environmental performance and literature provides limited empirical evidence on the effect of demographic board diversity (Cordeiro et al., 2020; Nadeem et al., 2020). The relationship between board diversity and environmental performance in family firms is under explored despite family firms constituting the backbone of most economies across the world (Anderson and Reeb, 2003): they produce 70–90% of gross domestic product around the world (De Massis *et al.*, 2018) and board characteristics are a relevant source of heterogeneity among them (Daspit *et al.*, 2021).

Drawing on different and competing theoretical perspectives, research has formulated arguments for a positive or negative relationship between family firms and environmental performance while empirical studies have also failed to provide unambiguous results regarding the direction of the family effect on environmental performance (Miroshnychenko *et al.*, 2022). On the one hand, some studies reveal a positive association between family firm status and environmental performance (Berrone *et al.*, 2010; Gomez-Mejia *et al.*, 2019; López-González *et al.*, 2019). On the other hand, some contributions highlight a negative association (Dal Maso *et al.*, 2020) or do not find a significant relationship between family firms and environmental performance (Cruz *et al.*, 2014). Therefore, it is relevant to address the issue of

how governance characteristics, other than ownership structure, may affect family firms' environmental performance.

Our research questions are: How does board diversity affect family firms' environmental performance? Does board tenure moderate the effect?

We address the questions by analyzing a sample of French, German, Italian, Portuguese and Spanish non-financial listed firms for the period 2014–2021. The sample is of interest because of the lack of studies on board diversity and environmental performance in family firms in European countries. Family firms have been playing a major role in continental European economies for centuries, particularly in the countries we focus on, where they reach the highest percentage of companies [1].

This paper draws on Agency Theory and Stewardship Theory, integrated by the Socioemotional wealth (SEW) perspective, as the aim of the paper is to analyze the effect of board diversity on environmental performance in family and non-family companies, as well as within family businesses, in order to highlight possible sources of heterogeneity among family companies' environmental performance. We articulate board diversity into structural diversity (proxied by board independence) and demographic diversity (proxied by gender), consistent with Veltri *et al.* (2021). Focusing on family firms, we add the proportion of family members on the board as a further demographic diversity variable. We also analyze the possible moderating effect of board tenure on the relationship between the above-mentioned forms of board diversity and environmental performance. We control for board size, as well as for other firm characteristics related to profitability, capital structure and socioemotional wealth, such as the owner's identity in the company name, family CEO and family ownership stake.

Our findings contribute to extant literature in different ways.

We contribute to environmental performance studies by pointing out that, in family firms, environmental performance is affected only by demographic diversity whilst environmental performance is significantly affected by structural board diversity, in terms of board independence and board size, as well as by gender diversity in non-family firms. This is the first study to analyze the relationship between structural board diversity and environmental performance in family firms and it is the first piece of research to provide empirical evidence on the association between gender diversity and family firm's environmental performance in a European setting.

We add to family firms' literature as we find that gender diversity has a positive and more significant impact in family firms than in non-family firms, but that the effect tends to diminish as board tenure increases. Moreover, within family firms, family members on the board tend to decrease environmental performance and the effect is higher for long tenured boards.

We contribute to board diversity literature as, to the best of our knowledge, ours is the first study to provide evidence of a different effect of board tenure in family and non-family firms. Our results suggest that, in family companies, the effect of demographic board diversity on environmental performance is moderated by board tenure, but that this effect does not hold for non-family companies.

The paper continues as follows: section 2 explains the theoretical framework; section 3 introduces the literature review and develops hypotheses; section 4 presents data and methods; section 5 presents and discusses the results; section 6 concludes the paper, highlighting the scientific contribution, implications for theory and practice, limitations of the research and avenues for further research.

2. Theoretical framework

Two competitive perspectives inform family firm environmental performance studies, leading to mixed results (Miroshnychenko *et al.*, 2022). The positive or negative valence of

JFBM

family control and influence is related to an owning family agency or stewardship view (Le Breton-Miller and Miller, 2016).

According to Agency theory, a company's commitment to activities favoring stakeholders is linked to three types of agency conflict. The first type relates to the conflict between owners and managers, nevertheless family businesses present a concentrated ownership structure that increases by monitoring incentives as well as capabilities, and reduces agency conflicts between owners and managers (Shleifer and Vishny, 1986; Anderson and Reeb, 2003). Moreover, an owning family may exert an indirect as well as a direct influence on management through family members' involvement in management. In this case, family managers may be risk-averse to the adoption of sustainability initiatives as they bear the residual risk of management decisions (Ernst *et al.*, 2022), to the detriment of environmental performance.

The second type of agency conflict arises between majority and minority shareholders. According to literature, family firms experience more severe second type agency conflicts than non-family businesses as family blockholders may extract private benefits from the business (Villalonga and Amit, 2006). Second type agency conflicts may negatively affect environmental performance as they arise from a family-centered view of the business as well as from the family's wealth concentration in the firm (Purkayastha et al., 2022). Noteworthy is the example of Zara, the first fast fashion company in the world, managed under a familycentered model, which shows no sign of slowing down its fast business and diminishing the high carbon footprint of its supply chains [2]. Namely, the presence of strong family ties may result in the prioritization of family members' needs, nepotism, entrenchment and the appointment of unskilled family managers to the detriment of organizational performance and of other stakeholders' interests (Schulze et al., 2001). Negative familism may demotivate non-family employees with negative effects on their performance and on their commitment to activities in support of the environment when these are not directly related to their duties (Sendlhofer, 2020). Agency conflicts of the second type may result in poor performance and, as a consequence, in the limited availability of resources to support environmental activities (Miroshnychenko et al., 2022). Moreover, family blockholders present under-diversified portfolios and a large part of a family wealth is invested in the business. For this reason, they are less likely to invest in uncertain projects (Gomez-Mejia et al., 2007) such as proenvironment projects, they are less prone to engage in social and environmental investments as most of the cost falls onto the owning family (Barnea and Rubin, 2010).

A third type of agency conflict emerges between shareholders and lenders (Mauer and Sarkar, 2005). On the one hand, shareholders may expropriate lenders wealth by the means of asset substitution operations (Jensen and Meckling, 1976). On the other hand, lenders tend to prevent this negative effect by demanding high remuneration and costly measures to monitor borrowers (Miroshnychenko *et al.*, 2022). There is evidence that family businesses prefer to finance growth with debt rather than external equity (Romano *et al.*, 2001) in order to maintain tight control, therefore costs related to agency conflicts with lenders may reduce the resources available for pro-environmental practices.

According to Agency theory, family businesses would be less prone to engage in environmental activities because of the peculiarities of their agency conflicts. Effective boards may limit agency conflicts by means of their monitoring capabilities (Jensen and Meckling, 1976), improve openness to change (Haniffa and Cooke, 2002) and provide resources to enhance environmental performance (De Villiers *et al.*, 2011). Board effectiveness and interest in environmental activities are affected by board characteristics (García Martín and Herrero, 2020) and, from an Agency perspective, effective boards need to be composed of independent directors (Rhoades *et al.*, 2000) and demographic diversity should enhance board monitoring because of the resultant less family-oriented system of values and thinking.

From the point of view of tenure, long-tenured directors may continue in the role because they are cronies of the owning family or management, their decisions are aligned with the interests of the owners and management, therefore board long tenure does not reduce agency] conflicts (Le-Breton-Miller *et al.*, 2015). From an agency perspective, a long average tenure may moderate the positive effect of board independence and gender diversity on environmental performance as independence tends to be only formal and the board is less prone to change the status quo and leave room for diverse thinking and innovation (Westphal and Zajac, 1995).

Stewardship theory may be seen as a particular case of Agency theory in which the principal and the agents have similar goals, thus minimizing first type agency conflicts (Caers *et al.*, 2006). According to Stewardship theory, many leaders and executives are driven by more than economic self-interest in their jobs: they often behave with altruism towards the organization and its stakeholders, and these attitudes would be prevalent among family companies whose leaders are either family members or who are emotionally linked to the owning family (Miller and Le Breton-Miller, 2006). From the Stewardship perspective, family firms' managers behave like good stewards as they aim to preserve the business for the family and for subsequent generations (Le Breton-Miller and Miller, 2006). A long-term orientation together with a strong family commitment to the continuity of the family, and to the firm's identity, results in family ownership having a positive effect on the adoption of proactive environmental strategies as an ethical form of behavior (Dou *et al.*, 2019) that creates a long-term business sustainability. Barilla, a century-old family business operating in the food sector, offers an example of this behavior through its long-lasting commitment in terms of sustainable packaging, raw materials and regenerative agriculture (Ruini *et al.*, 2016).

According to the Stewardship perspective, family firms are interested in long-term sustainability and, for this reason, they are prone to invest in CSR activities in order to straighten relationships with their internal and external stakeholders such as non-family employees, customers, suppliers and the environment (Le Breton-Miller and Miller, 2016). Family CEOs rely on a long tenure and later generations' care implies a long-term perspective. These governance characteristics result in lengthy investment time horizons and incentives to invest that may favor environmental activities that do not lead to a short run return but that improve environmental performance. Under this perspective, the role of the board is to support management in pursuing a company's goals, therefore a firm's board and management act as a single stewardship team for the business.

Unlike the Agency perspective, Stewardship Theory argues that internal directors are more beneficial to the firm (Donaldson, 1990; Donaldson and Davis, 1994). From this view, internal directors are good stewards of the company's interests, they will behave more diligently to preserve the business for future generations and invest in a longer-term perspective than independent directors (Baysinger and Hoskisson, 1990; Hill and Snell, 1988) and thus they may be more prone to invest in environmental activities. Under a Stewardship lens, a longer average tenure increases independent directors' experience and commitment to the firm in a longer-term perspective (Vafeas, 2003) as over time directors develop social ties that foster the identification with the firm (Sundaramurthy and Lewis, 2003) to the benefit of environmental engagement. Board demographic diversity can enhance stewardship by improving a firm's decision-making process by means of diverse views, experiences and sensitiveness (Nguyen et al., 2021), thus broadening stewardship to benefit a wider range of stakeholders and the environment. Under this view, a longer board tenure reduces board's discussions and may uniform views and opinions inside the board (Davis et al., 2007), moderating the positive effect of demographic diversity on environmental performance. In family firms this implies an alignment to the owning family priorities and translates into a lower propensity to entrepreneurial risk (Huybrechts et al., 2013) and the pursuit of environmentally risky strategies.

Stewardship behavior and agency conflicts in family firms are affected by the relevance non-financial goals assume in this type of company (Chrisman, 2019). Under the Socioemotional wealth (SEW) perspective, family firms' behavior is led by non-financial

JFBM

goals that consist in the preservation of the systems of emotional values a family extracts from the business by the means of ownership control (Gomez-Mejia *et al.*, 2007). A number of dimensions pertain to the SEW concept. Family control and influence on the business provide the family with visibility and reputation; family members feel a relevant sense of identification with the firm that projects a strong family image in society; a family builds social ties through the business deriving status and reputation; dynastic succession renews family bonds to the business (Berrone *et al.*, 2012).

On the one hand, the identification of the family with the firm and the desire to ensure its longterm survival may increase a propensity towards environmentally sustainable practices (Berrone *et al.*, 2010; Agostino and Ruberto, 2021). On the other hand, SEW preservation may result in a family centered decision-making process to the detriment of the environment and to a firm's stakeholders at large (Kellermanns *et al.*, 2012, p. 1179). Family control and influence over the firm may reinforce risk aversion and result in dysfunctional conservatism (Miller and Le Breton-Miller, 2014), reducing the propensity to engage in environmentally friendly initiatives. In this vein, family ownership and control negatively affect the adoption of green product innovation, which may put at risk family assets (Huang *et al.*, 2016). Board diversity may moderate the possible negative effect of SEW on environmental performance, bringing different and less family-centered values and sensitivities onto the board (Gavana *et al.*, 2023).

3. Literature review and hypotheses development

3.1 Independent directors and environmental performance

According to Stewardship theory, internal directors are better stewards of a business than independent directors and they invest in a longer-term sustainability perspective (Hill and Snell, 1988; Baysinger and Hoskisson, 1990), which may benefit environmental investments and by its means environmental performance. Nevertheless, only a limited number of studies points out a negative effect of independent directors on environmental performance (Walls et al., 2012; Orazalin and Mahmood, 2021). According to Agency theory, a higher proportion of independent directors can monitor the opportunistic behavior of managers more effectively (Luan and Tang, 2007). Independent directors feel less pressure toward financial goals and exhibit greater concerns for CSR (Ibrahim and Angelidis, 1995), promoting the balance between financial and environmental objectives and between short-term and longterm goals (Liao *et al.*, 2015). They are more prone to use their competencies and experiences to improve environmental performance in order to strengthen their relationship with stakeholders and achieve reputational rewards (Mallin et al., 2013). Accordingly, a majority of empirical studies show that increasing the number of independent directors positively contributes to environmental performance (De Villers *et al.*, 2011; Post *et al.*, 2011, 2015). A greater number of independent directors helps the board to formulate and implement a proactive and comprehensive CSR strategy and, in turn, to achieve higher environmental performance (Shaukat et al., 2016). None of the previous studies have analyzed the effect of independent directors controlling for the nature of the firm, in terms of family or non-family ownership control. Nevertheless, the literature suggests that family business status is a specific company condition that may influence the relationship between board diversity and non-financial performance (Veltri et al., 2021). There is evidence that family firms' independent directors tend to be related to the controlling family, and in so doing, they are less effective in lowering agency conflicts (Chen and Jaggi, 2000) that may jeopardize environmental performance. Moreover, according to literature, family firms' CSR behavior is guided not only by financial objectives but also by the pursuit of socio-emotional wealth. Under an Agency perspective, SEW reduces agency conflicts (Gavana et al., 2022) and may lower the beneficial effect of independent directors on environmental performance. From a Stewardship view, SEW provides internal directors with a long-term perspective (Berrone

et al.. 2012) suitable for environmental investment that would result in a lower effect of Effect of board independent directors on environmental performance in family than in non-family firms.

H1. The beneficial effect of independent directors on environmental performance is lower in family than in non-family firms

3.2 CEO non-duality and environmental performance

Under an Agency perspective, CEO duality can reduce board independence, blur the borderline between management and control (Luan and Tang, 2007) as well as undermine the effectiveness of board monitoring (Mallin et al., 2013), whilst non-dual leadership structures are more prone to pursue with diligence their monitoring role. On the one hand, dual CEOs may be more profit-oriented than non-dual CEOs and the latter would benefit social and environmental activities (Zhang, 2012). Accordingly, certain empirical studies have shown that CEO duality is negatively associated with environmental performance (De Villers et al., 2011; Hussain et al., 2018). On the other hand, more powerful CEOs of firms operating in polluting industries may increase environmental performance to extract resources from the business, as good environmental performance enhances CEO pay (Berrone and Gomez-Mejia, 2009) and CEO non-duality would limit this opportunistic behavior. Under an Agency perspective, we would expect a stronger effect of CEO non-duality for non-family firms as agency conflicts of the first type are tougher than in family firms.

Under a stewardship perspective, managers are good stewards of a business, they operate in order to ensure a firm's long-term sustainability. Donaldson and Davis argue that the overlapping position of board chair and CEO favors a CEO's steward commitment (Donaldson and Davis, 1991). CEO duality favors strong leadership and effective strategic decision-making (Hewa Wellalage and Locke, 2011) also to the benefit of the environment. Consistently, Arena et al. (2015) find a positive relationship between CEO duality and environmental performance and Walls et al. (2012) point out that environmental performance decreases in absence of CEO duality. Therefore, CEO non-duality would be less effective in favoring environmental performance. There is evidence that some CEO characteristics differently affect family and non-family firms' environmental performance because of the relevance of the SEW and its beneficial effect for the environment (Berrone *et al.*, 2010). Splitting the authority between the board chair and the CEO may reduce family influence (Cabrera-Suárez and Martín-Santana, 2015). Nevertheless, on the one hand, family CEOs are led by non-financial goals (Blanzo-Mazagatos et al., 2022) and, on the other hand, non-family CEOs tend to develop over time a sense of psychological ownership and share family goals and values (Huybrechts et al., 2013). Therefore, we expect that the possible negative effect of CEO non-duality on environmental performance is lower in family than in non-family firms because of the beneficial effect of the SEW.

H2. The effect of CEO non-duality on environmental performance is lower in family than in non-family firms

3.3 Women on the board and environmental performance

Regarding boards' demographic characteristics, literature has addressed the effect of board gender diversity on environmental performance, reaching mixed results. Under an Agency theory perspective, female directors may reduce agency conflicts to the benefit of CSR performance because of their enhanced monitoring capabilities and sensitiveness (Veltri et al., 2021) unless their appointment derives from formal criteria, not allowing them to affect strategic decision-making (Fernandez-Feijoo *et al.*, 2014). According to Stewardship theory, board members support a firm's management rather than monitor its behavior (Corbetta and Salvato, 2004). Directors and managers act together in favor of the long-term sustainability of

diversity and tenure

the business and the presence of women favors stewards' behavior towards other stakeholders and the environment.

Several empirical studies have revealed a positive association between the presence of women on the board and environmental performance. Women bring to the board different skills and resources, and this benefits environmental performance, especially in more polluting industries where there is a higher complexity in managing firms' environmental impact (Li et al., 2017; Lu and Herremans, 2019). Women enrich the board with different values, ideas, knowledge and perspectives. They exhibit diverse thinking styles, are more prone than men to meet the needs of multiple stakeholders and present a long-term orientation that enhances environmental performance (Nadeem et al., 2020; Orazalin and Mahmood, 2021; Islam et al., 2022). Women care for others more than men and, among a firm's stakeholders, female directors tend to prioritize less powerful stakeholders, such as the environment (Francoeur et al., 2019). The educational background of female directors, more oriented towards humanities and education, as well as their different professional experiences, help the board to improve a firm's environmental performance (García Martín and Herrero, 2020). Increasing the number of women on the board raises the likelihood of forming sustainability-themed alliances beneficial for environmental performance (Post *et al.*, 2015), even if research points out that women are able to positively influence a firm's environmental performance when reaching a critical mass on the board (Post et al., 2011). A limited number of contributions finds a non-significant or a weak effect of women on the board on environmental performance (Walls et al., 2012; Hussain et al., 2018). Family firms are characterized by a long-term perspective as their decision-making process is led by nonfinancial goals, such as maintaining family influence on the business in order to pass the business on to future generations (Berrone *et al.*, 2012). They are particularly concerned about a firm's reputation and image risk (Santos et al., 2022), as a company's public image affects the SEW endowment derived from the business (Razzak and Jassem, 2019). For this reason, they are very attentive to the judgment of the community in which they operate and they try to satisfy stakeholders' needs related to environment preservation (Berrone et al., 2010). Women enrich the board with greater sensitivity towards stakeholder needs; family owners are aware of this and rely on female directors to implement their environmental CSR agenda (Cordeiro *et al.*, 2020). Based on this discussion, we expect that an increase in the ratio of female directors on the board would be particularly effective in enhancing environmental performance in family firms.

H3. The positive effect of women ratio on the board is higher in family than in non-family firms

3.4 Family members on the board and environmental performance

Family firms may behave differently towards environmental performance depending on the presence of family directors on the board (López-González *et al.*, 2019), a further and peculiar source of demographic board diversity for family businesses (Gavana *et al.*, 2023). Family involvement in management may influence negatively or positively strategic decisions, such as engagement in social and environmental activities. Under an Agency theory perspective, the presence of the family on the board increases second type agency conflicts and may lead to a family-centered decision making-process, to the detriment of minority shareholders as well as external stakeholders (Le Breton-Miller and Miller, 2016). Moreover, family involvement may exacerbate conflicts among family members, especially when they belong to different branches of the family. These branches may strive for different goals, causing disagreement on the firm's strategic orientation (Eddleston and Kellermanns, 2007) to the detriment of environmental performance. Under a Stewardship view, family business leaders are particularly motivated to serve as stewards for the collective good as they belong, or are linked to, the owning family (Miller and Le Breton-Miller, 2006). The presence of family members on the board may support a firm's management in pursuing a firm's long-term

sustainability and in engaging in investments characterized by long-term returns such as environmental activities. Family involvement in management may result in a higher sense of identification of the family with the firm and prompt family members to undertake actions, such as CSR initiatives, that favor the firm's reputation and image (Khojastehpour and Johns, 2014; Kuttner *et al.*, 2021), contributing to SEW preservation and the firm's survival (Sageder *et al.*, 2018). Empirical research provides evidence that a larger presence of family members on the board further boosts social and environmental performance (López-González *et al.*, 2019). Conversely, the study by Graafland (2020) shows that environmental performance is higher for family firms managed by a balanced mix of family and non-family members. This is because family firms fully managed by family members may prioritize the interests of the family over those of other stakeholders, whereas the absence of family managers reduces identification with the firm and, in turn, the importance of the firm's image and reputation.

H4. The ratio of family members on the board affects family firms' environmental performance

3.5 Board tenure and environmental performance

Under a Stewardship perspective, a longer board tenure can support management in improving a firm's environmental performance, particularly in family firms due to their longterm sustainability perspective. Directors with longer tenure are more familiar with the firm's operations (Fischer and Pollock, 2004). This in-depth and specific knowledge enables them to better assess risks and seize opportunities stemming from the environmental challenges that the company should face (Katmon et al., 2019) and better allocate resources to meet stakeholders' need, with a positive effect on environmental performance (Paolone et al., 2023). From an Agency theory perspective, low director turnover may result in entrenchment and lower monitoring capabilities as long-tenured directors may be left in place because they are linked to the management and to the owning family (Le Breton Miller et al., 2015). Moreover, literature points out that a long board tenure may result in directors developing close friendships with managers and loosening monitoring over them. Further, the lack of director turnover may prevent new perspectives and fresh ideas from reaching the board (Vafeas, 2003), slowing down innovation in social and environmental activities (Patro et al., 2018). This implies that a long tenure may reduce monitoring effectiveness of board independence and its influence on environmental performance as it inhibits diverse thinking that is beneficial to environmental performance (Elmagrhi et al., 2019). There is evidence that non-family CEOs' long tenure levels out entrepreneurial risk-taking because of an alignment with the owning family' goals and perspective (Huybrechts et al., 2013), and longer tenured boards may operate in a family-centered perspective. Literature points out that a low turnover may slow down innovation in family firms as it results in high groupthink, limiting openness to external knowledge and to strategic changes (Rondi et al., 2021) such as those implied by environmental challenges. This suggests that a long tenure may limit the beneficial effect of board demographic diversity on environmental performance because of an alignment of values and perspectives among board members. Board diversity improves environmental performance as different perspectives and values help the board to evaluate the needs and expectations of different stakeholders (Islam et al., 2022). Long tenure may increase directors' risk aversion and *inertia* to change (Golden and Zajac, 2001) and develop groupthink inside the board (Forbes and Milliken, 1999). In particular, family directors' long tenure can create reluctance in adopting practices that improve environmental performance as they bring a family-oriented view to the board for a long period of time (Le Breton-Miller and Miller, 2016).

H5. Board tenure differently affects the relationship between board diversity and environmental performance in family and non-family firms and within family firms.

4. Data and methods **IFBM**

4.1 Data collection

The data sample includes non-financial firms listed on the French, German, Italian, Portugal and Spanish exchanges (Euronext Paris, Deutsche Börse, Euronext Milan, Euronext Lisbon, Bolsa de Madrid) in the period 2014–2021. French, German, Italian, Portugal and Spanish are civil-law countries with comparable legal settings and similar level of financial market development. These countries are characterized by a strong presence of closely family-controlled companies (Franks et al., 2012; Faccio and Lang, 2002). Before the issuing of the EU Non-Financial Reporting Directive (2014) a limited number of companies disclosed their non-financial performance and we focused the analyses on the period 2014-2021 in order to maximize data availability in the Refinitiv Eikon database, which elaborates information retrieved from companies' sustainability reports. The initial sample comprised firms with available Environmental performance pillar score (ENVP) in the Refinitiv Eikon database. After removing observations with missing data the final sample consisted, on average, of an unbalanced panel of 434 firms. We define a family company as one where a family is the ultimate controlling owner, with a minimum control threshold of 20%. Other research, focused on the U.S., categorizes a company as a family firm when "the founder or at least one member of his or her family acts as a director or an executive officer and the founding family owns at least 5%" (Martins and Pires, 2023). Our threshold is consistent with those used in research focused on European listed companies, as these countries are characterized by a higher ownership concentration than the U.S. (Ellul, 2010; Faccio and Lang, 2002). Consistently with literature focused on listed companies (Croci et al., 2011) we use an ownership-based definition as it allows us to point out the effect of different levels of family involvement based on ownership control, percentage of family members on the board, presence of a family CEO and company-family name identity on environmental performance. In doing so, we end up with a final sample of 171 family firms and 263 non-family firms and 1897 observations.

We hand-collected information on ownership and board characteristics from companies' corporate governance reports. The balance-sheet data comes from Orbis, the global Bureau van Dijk database. The Environmental performance pillar score derives from the Refinitiv Eikon database.

4.2 Variables

Environmental performance measures the impacts of an organization's environmental policies. We proxy the environmental performance with the firm's score calculated annually in the Refinitiv Eikon database. Refinitiv Environmental Pillar Score is a proxy to measure the level of environmental performance and it has already been tested in recent scientific research (Paolone et al., 2023; Orazalin and Mahmood, 2021; Guérin and Suntheim, 2021; García Martín and Herrero, 2020; Nadeem et al., 2020). KLD ratings, based on 14 environmental "strength" and "concerns" variables, have been used in less recent studies on the link between corporate governance and social performance (Walls et al., 2012). Nowadays, the Refinitiv Eikon Datastream is the leading database for environmental, social and governance data worldwide (Shakil et al., 2022). The environmental global score gathers information on 68 metrics grouped in three categories related to resource use, emissions and innovation (Nadeem et al., 2020; Refinitiv, 2022). The first category refers to the reduction in water, energy, land and air consumption due to more eco-efficient solutions. The second category relates to the reduction of environmental emissions in business operations and the third category to the reduction of environmental costs for customers by means of new environmental technologies and processes. The Environmental Pillar Score bases on 20 indicators of resource use, 28 indicators of emissions and 20 indicators of innovation (Refinitiv Eikon Datastream, 2022).

Table 1 reports the dependent, independent and control variables used in this study. IBD, NCD, Effect of board BS, BT proxy for structural board diversity in terms of weight of independent directors, CEO nonduality, board size and board tenure. Boards of directors may differ in their level of independence. Two of the main characteristics that express board independence are the proportion of independent directors and the lack of overlap between the CEO and the board chairman (Zhang, 2012). Independent directors can have a positive effect on firm's environmental performance (Post et al., 2015). CEO non-duality may affect board independence and environmental performance

diversity and tenure

Variable type	Variable	Measurement	Measure of	Source
Dependent variable	ENVP	Environmental performance pillar score	Environmental performance	Refinitiv Eikon
Independent variables	IBD	Number of independent directors divided to the number of directors on the board	Diversity of the board	Corporate governance reports
	NCD	A dummy variable that coded 1 if the CEO is not also the chairperson of the board of directors	Diversity of the board	Corporate governance reports
	WBD	Ratio of female director to total number of directors	Diversity on the board	Corporate governance reports
	FBD	Ratio of family directors to total number of directors	Diversity on the board	Corporate governance reports
Moderator Control variables	BT BS	Average board tenure Number of board members	Diversity of the board Other board structure characteristic	Refinitiv Eikon Corporate governance reports
	FCEO	A dummy variable coded 1 when the CEO belongs to a family that owns at least 20% of the firm's common shares	Family influence	Corporate governance reports, companies websites
	FOWN	Sum of the controlling family's equity stakes (%)	Family influence and capital structure	ORBIS, Consob website
	ID	A dummy variable coded 1 when the name of the family or controlling shareholders appears in the company name	Family/controlling shareholder's identification with the firm	Hand-collected from companies websites
	CEOAGE	Age of the CEO in years	CEO characteristic	ORBIS, companies websites
	CEOB	A dummy variable coded 1 when the CEO is a member of the board	CEO characteristic	Corporate governance reports, companies websites
	LEV	Ratio of total financial debt divided by equity	Capital structure	ORBIS
	SIZE AGE ROE	Log of total assets Age of the firm in years Return on equity, ratio of operating income divided by equity	Firm's visibility Firm's visibility Profitability	ORBIS ORBIS ORBIS
Source(s): Au	thors own cro	1 0		

Table 1. escription of variables (Hussain *et al.*, 2018). Board size is the number of board directors, board tenure is the average board tenure of all board members of a given firm (Paolone *et al.*, 2023).

WBD and FBD proxy for demographic board diversity in terms of weight of female directors on the board and proportion of family members on the board, both proxies may affect the firm's environmental performance (Dal Maso *et al.*, 2020; Li *et al.*, 2017; López-González *et al.*, 2019; Nadeem *et al.*, 2020; Orazalin and Mahmood, 2021).

FCEO controls for the presence of a family CEO, while FOWN measures family ownership. We used the identity of the firm's and first owner names to construct an identification proxy (ID). We also control for leverage (LEV), performance (ROE) and for firms' log of assets (SIZE) and age (AGE). Finally, we defined dummy variables to take into account the possible effects of industry, year and country.

4.3 Model and methods

The basic model we estimate to analyze the impact of demographic and structural board diversity on environmental performance for family and non-family firms is the following:

$$ENVP = \alpha_{0} + \beta_{1}IBD + \beta_{2}NCD + \beta_{3}WBD + \beta_{4}BS + \beta_{5}BT + \beta_{6}ID + \beta_{7}ROE + \beta_{8}LEV + \beta_{9}SIZE + \beta_{10}AGE + \sum_{j=1}^{j}\gamma_{j}I_{j,it} + \sum_{t=1}^{T}\delta_{t}D_{t} + \sum_{h=1}^{H}\varphi_{h}C_{h,it}$$
(1)

With model (1) we measure the effect of structural and demographic board diversity for family and non-family firms controlling for the influence of financial performance, leverage, size and firm age.

We also estimate an augmented model to verify the moderating effect of board tenure on the relationship between board diversity and environmental performance. Specifically, we verify hypothesis H5 estimating the interaction effects of BT with NCD, IND and WBD:

$$ENVP = \alpha_0 + \beta_1 IBD + \beta_2 NCD + \beta_3 WBD + \beta_4 BS + \beta_5 BT + \beta_6 ID + \beta_7 ROE + \beta_8 LEV + \beta_9 SIZE + \beta_{10} AGE + \sum_{n=1}^{N} BTinteractions + \sum_{j=1}^{j} \gamma_j I_{j,it} + \sum_{t=1}^{T} \delta_t D_t + \sum_{h=1}^{H} \varphi_h C_{h,it}$$
(2)

Finally, to verify the robustness of the results for family firms, we introduce specific governance variables in order to control for the presence of family members on the board, a family CEO and for family ownership:

$$ENVP = \alpha_{0} + \beta_{1}IBD + \beta_{2}NCD + \beta_{3}WBD + \beta_{4}BS + \beta_{5}BT + \beta_{6}ID + \beta_{7}ROE + \beta_{8}LEV + \beta_{9}SIZE + \beta_{10}AGE + \beta_{11}FBD + \beta_{12}FCEO + \beta_{13}FOWN + \sum_{j=1}^{j}\gamma_{j}I_{j,it} + \sum_{t=1}^{T}\delta_{t}D_{t} + \sum_{h=1}^{H}\varphi_{h}C_{h,it}$$
(3)

The last model permit us to verify the moderating effect of board tenure on the relationship between board diversity and environmental performance in family firms controlling for the presence of family members on the board, a family CEO and for family ownership:

$$ENVP = \alpha_0 + \beta_1 IBD + \beta_2 NCD + \beta_3 WBD + \beta_4 BS + \beta_5 BT + \beta_6 ID + \beta_7 ROE + \beta_8 LEV$$

$$+ \beta_9 SIZE + \beta_{10} AGE + \beta_{11} FBD + \beta_{12} FCEO + \beta_{13} FOWN + \sum_{n=1}^{N} BTinteractions$$

$$+ \sum_{j=1}^{j} \gamma_j I_{j,it} + \sum_{t=1}^{T} \delta_t D_t + \sum_{h=1}^{H} \varphi_h C_{h,it}$$
(4)

To take into account the impact on the results of the presence of firms of different countries, industries and years, we control for fixed effects along these three dimensions. The dependent variable in the above models is censored assuming values from 0 to 100, we estimate models (1)–(4) using Tobit regression models separately for family and non-family firms, and control for time, country and industry fixed effects using dummies (Gao and Connors, 2011; Nguyen *et al.*, 2021).

5. Results and discussion

5.1 Results

In Table 2 we present the descriptive statistics for environmental performance, explanatory and control variables. The environmental performance score varied from 0 (minimum) to 99.20 (maximum), with a mean of 56.90 and a median score of 61.69. Board tenure ranged from less than 1 year to 33 years. The separate analysis of family and non-family firms shows significant differences in the means of most of the variables at the one percent level of the *t*-statistics.

Table 3 reports the Pearson correlation coefficients, most of the correlations among variables are low or moderate. The only variable that shows a high correlation is the ratio of family directors on board (FBD) with the family CEO dummy (FCEO), but this is reasonable. Furthermore, the correlations between explanatory variables as well as the VIF factors are relatively small, and we can rule out multicollinearity effects on the reliability of the estimated parameters. Influence diagnostics indicated no problematic outliers in the data sample.

Table 4 presents of the impact of board structural and demographic diversity on firms' environmental performance, taking into account the possible moderating effect of board tenure. In particular, columns 1 and 2 respectively report the results of model (1), testing the first three hypotheses (H1-H3) relatives to the impact of independent directors, CEO non-duality and women presence on the board for non-family and family firms. Columns 3 and 4 show the results of model (2), testing with the same set of variables the moderating effect of board tenure for non-family and family firms (hypothesis H5). Finally, columns 5 and 6 display the results of models (3) and (4) for family firms, regarding the impact of family members on the board and the interaction with board tenure (hypotheses H4-H5).

To verify the robustness of these results we also added some other control variables to capture the impact of CEO characteristics and of the possible existence of an inverted U-shaped relationship between environmental performance and family ownership. The results of these additional analyses are tabulated in Table 5. CEO age as well as CEOB - a dummy to control if the CEO is also member of the board - positively affect family firm's environmental performance whilst an inverted U-shaped relationships relates environmental performance to family ownership. After controlling for these variables the results on the effect of structural and demographic board diversity remain qualitatively the same.

5.2 Discussion

Our findings point out that environmental performance is significantly lower in family than in non-family firms. This result is consistent with an Agency perspective, under which the

JFBM		All firms	Non-FF	FF	<i>t</i> -test
	ENVP	56.90 (26.46)	58.22 (26.92)	54.57 (25.48)	3.12 ***
	IBD	0.41 (0.19)	0.44 (0.19)	0.36 (0.18)	12.24***
	NCD	0.79 (0.41)	0.83 (0.38)	0.72 (0.45)	8.02***
	BS	11.89 (4.87)	11.89 (5.09)	11.89 (4.52)	0.02
	WBD	0.30 (0.15)	0.30 (0.15)	0.30 (0.15)	-0.61
	FBD	0.09 (0.15)		0.21 (0.16)	
	 BT 	6.82 (3.29)	5.97 (2.80)	8.18 (3.54)	-19.69^{***}
	CEOAGE	57.67 (7.93)	57.74 (8.06)	57.55 (7.69)	0.53
	CEOB	0.55 (0.50)	0.51 (0.50)	0.61 (0.49)	-4.45^{***}
	FCEO	0.26 (0.44)		0.47 (0.50)	
	FOWN	0.21 (0.27)		0.52 (0.16)	
	LEV	2.10 (2.86)	2.22 (3.22)	1.91 (2.17)	3.48***
	ROE	0.13 (0.19)	0.12 (0.20)	0.14 (0.16)	-4.76^{***}
	ID	0.11 (0.32)	0.02 (0.13)	0.26 (0.44)	-20.46^{***}
	SIZE	14.57 (2.09)	14.73 (2.25)	14.33 (1.77)	5.99***
	AGE	61.05 (49.04)	55.04 (47.76)	70.56 (49.55)	-9.47^{***}
	FIRMS	434	263	171	
Table 2.Descriptive statistics	• • • •	*, *: 1, 5 and 10% signific thors own creation	cance levels, respectively		

concentration of family wealth in a business makes family firms more risk averse than nonfamily companies as well as lowering their environmental engagement and performance (Dal Maso *et al.*, 2020). Environmental innovation in a firm's processes and products is risky (Zaman et al., 2023), and due to controlling family undiversified portfolios, family companies are less prone to engage in these strategies to the detriment of environmental performance. Our finding is also consistent with empirical evidence reporting a negative relationship between family blockholders and environmental performance (Dal Maso *et al.*, 2020; Forés et al., 2022). Our findings reveal that environmental performance in family firms varies according to the level of family ownership under an inverted U-shaped relationship. This suggests that family ownership tends to increase environmental performance because nonfinancial objectives related to reputation prevail (Berrone et al., 2010) on short-term financial goals. Conversely, beyond an optimum level of family ownership (52,2% of the share capital), uncertain future financial returns of innovation investments may jeopardize SEW preservation in the long run (Chrisman and Patel, 2012). The concern for financial sustainability and the desire to preserve family wealth for future generations (Geng et al., 2023) make family companies less prone to invest in risky activities (Mitter et al., 2014) to the detriment of environmental performance.

According to literature, board diversity may influence environmental performance via structural as well as demographic characteristics (García Martín and Herrero, 2020; Orazalin and Mahmood, 2021; Islam *et al.*, 2022). Structural diversity impacts on board monitoring role to the benefit of the firm's different stakeholders and demographic diversity widens the board's perspective towards more stakeholder-oriented strategies (Liao *et al.*, 2015; Elmagrhi *et al.*, 2019). We find that board diversity differently affects environmental performance in family and non-family firms. Non-family firms' environmental performance is affected by diversity of board structure as well as demographic board diversity, whilst family companies' environmental activities are significantly related only to the latter.

As for structural board diversity, we find that CEO non-duality negatively affects environmental performance, but that the effect is significant only for non-family companies, supporting H2. Under an Agency perspective, duality provides CEOs with great power, allowing

CEOB	$\begin{array}{c} 0.21\\ 0.17\\ -0.24\\ 0.34\\ 0.25\\ 0.25\\ 0.14\\ 0.14\\ 0.06\\ 0.00\\ 0.06\\ 0.09\\ 0.06\\ 0.09\\ 0.09\end{array}$	Effect of board diversity and
CEO AGE	$\begin{array}{c} 0.06\\ 0.12\\ 0.12\\ -0.08\\ 0.06\\ 0.06\\ 0.09\\ 0.01\\ -0.02\\ 0.01\\ 0.01\\ 0.01\\ 0.01\\ 0.01\\ 0.01\end{array}$	tenure
AGE	$\begin{array}{c} 0.16\\ -0.06\\ 0.17\\ 0.17\\ 0.02\\ 0.02\\ 0.02\\ 0.02\\ 0.02\\ 0.02\\ 0.02\\ 0.02\\ 0.02\end{array}$	
SIZE	$\begin{array}{c} 0.63\\ 0.22\\ 0.22\\ 0.37\\ 0.37\\ 0.37\\ 0.06\\ -0.04\\ -0.01\\ -0.01\\ 0.16\\ 0.16\end{array}$	
LEV	$\begin{array}{c} 0.12\\ 0.02\\ 0.02\\ 0.04\\ 0.06\\ -0.02\\ -0.02\\ -0.02\\ -0.02\\ -0.02\\ -0.02\end{array}$	
ROE	$\begin{array}{c} 0.02\\ 0.03\\ 0.05\\ 0.02\\ 0.04\\ 0.06\\ 0.06\\ 0.08\\$	
FOWN	$\begin{array}{c} -0.09 \\ -0.24 \\ -0.02 \\ -0.03 \\ 0.03 \\ 0.35 \\ 0.35 \\ 0.39 \\ 0.39 \\ 0.39 \end{array}$	
FCEO	-0.13 -0.22 -0.37 -0.37 -0.23 0.21 0.47	
FBD	-0.13 -0.24 -0.27 0.08 0.39 0.39	
BT	-0.06 -0.15 -0.13 -0.13 0.01 0.17	
D	-0.02 -0.05 0.11 -0.01 -0.01	
WBD	0.33 0.14 -0.23	
NCD	-0.13 0.07 0.12 0.12 s level on	
BS	ENVP 0.15 0.37 -0.1 BD -0.18 0.0 BS 0.18 NCD 0.0 WBD 0	
IBD	0.15 talic: 5% s	
	ENVP IBD BS NCD WBD DD BT FBD FCEO FOWN ROE FOWN ROE LEV SIZE AGE CEOAGE CEOAGE Note(s): II	Table 3. Correlation matrix

JFBM

	Model 1 NFF	Model 1 FF	Model 2 NFF	Model 2 FF	Model 3 FF	Model 4 FF
INT	-52.37 (6.98)***	-72.55(11.19)***	-44.71 (8.63)***	-100.79 (14.59)***	-66.63 (11.17)***	-96.57 (15.62)***
BD		5.62(4.59)	6.30(8.97)	10.40 (10.83)	-1.13(4.57)	7.53(12.02)
NCD		1.11 (1.89)	-0.62 (3.55)	4.03 (4.75)	0.58(2.10)	6.46(5.00)
WBD	5.51 (5.82)	$30.27 (6.65)^{***}$	-5.07 (12.81)	79.71(13.78)***	26.46 (6.54)*** 0 r0 /0 00)**	$52.98 (14.00)^{***}$
RT RT	1.52 (0.16)*** 0.69 (0.94)***	0.64 (0.23)*** 0.66 (0 21)***	1.52 (0.16)*** 1 82 (0.01)**	0.50 (0.23)** 9 05 (0 95)**	-0.41 (0.23)**	0.30 (0.24)** 9 07 (1 13)***
FBD		(177:0) 00:0			-43.41 (6.17)***	6.08 (17.08)
FCEO					2.04(1.86)	0.02 (1.88)
FOWN					-0.07 (4.98)	-1.18(5.02)
Ð	-3.87 (3.69)	-4.44 (1.64)***	-4.58(3.71)	$-5.16(1.64)^{***}$	-3.61 (1.66)**	-3.08(1.67)*
ROE	0.08 (0.03) **	0.03(0.04)	0.08 (0.03)***	0.03(0.04)	0.03 (0.04)	0.02 (0.04)
LEV	0.13(0.18)	$0.81 (0.31)^{***}$	0.13(0.18)	$0.75(0.31)^{**}$	$0.86(0.30)^{***}$	$0.83(0.230)^{***}$
SIZE	$4.71 (0.40)^{***}$	$7.63 (0.64)^{***}$	$4.71 (0.41)^{***}$	7.88 (0.64)***	7.27 (0.63)***	6.94 (0.65)***
AGE	$0.06(0.01)^{***}$	$0.04 (0.02)^{**}$	$0.05 (0.01)^{***}$	$0.04 (0.02)^{**}$	$0.04 (0.02)^{***}$	0.04 (0.02)**
Year	yes	yes	yes	yes	yes	yes
Industry	yes	yes	yes	yes	yes	yes
Country	yes	yes	yes	yes	yes	yes
BT*NCD			-0.26(0.52)	-0.29(0.51)		-0.80(0.54)
BT*IBD			2.08 (1.36)	-0.95(1.22)		-1.10(1.31)
BT*WBD RT*FRD			1.69(2.13)	-5.27 (1.36)***		-2.30 (1.44)
Rho	0.67	0.69	29.0	0.70	0.72	0.73
Log Likelihood	-5187	-2941	-5185	-2931	-2917	-2904
obs	1,208	689	1,208	689	689	689
Note(s): ***, **, *: 1, 5 and	: 1, 5 and 10% significan	10% significance levels, respectively				
Source(s): Authors own creation	s own creation					

Table 4.Tobit results

Model 4 FF	$\begin{array}{c} -147.16 \left(20.38 \right)^{***} \\ 13.77 \left(12.12 \right) \\ 3.93 \left(5.19 \right) \\ 63.50 \left(14.14 \right)^{***} \\ 0.10 \left(0.24 \right) \\ 2.80 \left(11.7 \right)^{**} \\ 0.01 \left(0.7.7 \right)^{**} \end{array}$	-2010 (11.31) 0.44 (0.09)*** 5.15 (2.93)* 0.09 (1.94) 144.50 (24.94)***	-2.11 (1.66) -2.11 (1.66) -0.01 (0.04) 1.13 (0.35)*** 8.61 (0.67)*** 0.01 (0.02)	yes	yes -0.27 (0.57) -2.33 (1.31)* -3.53 (1.44)** -3.31 (1.91)*	0.76 2737	655	Effect dive
Model 3 FF	$\begin{array}{c} -110.54 \ (14.86)^{***} \\ -4.73 \ (4.55) \\ 2.46 \ (2.04) \\ 2.947 \ (6.60)^{****} \\ 0.11 \ (0.23) \\ -0.57 \ (0.21)^{****} \\ c_0 \ 0.16 \ (2.5)^{****} \\ c_0 \ 0.16 \ (2.5)^{****} \end{array}$			yes yes	SS A	0.75 2745	655	
Model 2 FF	-99.69 (18.25)*** 19.78 (11.07)* -1.40 (4.97) 75.95 (14.50)*** 0.43 (0.24)* 1.79 (0.99)*	0.34 (0.10)*** 4.71 (3.06)	-4.48 (1.67)*** 0.02 (0.04) 0.86 (0.37)** 8.51 (0.65)*** 0.02 (0.02)	yes yes	$_{-4.97}^{yes}$ 0.45 (0.54) -2.18 (1.24)* -4.97 (1.43)***	0.72 -2780 34 (<0.0001) 46 (<0.0001)	655	
Model 2 NFF	-37.22 (9.22)**** 8.56 (9.01) -1.32 (3.57) -7.32 (13.02) 1.50 (0.16)**** -1.89 (0.91)**	-0.16 (0.07)** -0.60 (1.90)	-5.18 (3.73) 0.09 (0.03)**** 0.08 (0.18) 4.84 (0.41)**** 0.05 (0.01)****	yes	$^{\rm yes}_{-0.18(0.52)}$ 1.85(1.36) 2.09(2.16)	0.68 5143	1,199	
Model 1 FF	-66.19 (13.64)*** 4.82 (4.70) 2.13 (1.92) 30.21 (6.89)*** 0.59 (0.24)** -0.86 (0.22)***	0.29 (0.10)*** 4.41 (3.07)	-3.59 (1.67)** 0.02 (0.04) 0.71 (0.37)* 8.16 (0.65)*** 0.02 (0.02)	yes	Ycs	0.71 -2789 28 (<0.0001) 38 (<0.0001)	655 e levels, respectively	
Model 1 NFF	-45.26 (7.65)*** 20.16 (3.59)*** -2.59 (1.49)* 5.32 (5.88) 1.51 (0.16)*** -0.60 (0.24)**	-0.16 (0.07)** -0.70 (1.89)	-4.54 (3.71) 0.09 (0.03)**** 0.08 (0.18) 4.83 (0.41)**** 0.06 (0.01)****	yes	Ycs	0.68 5145	obs 1,199 655 Note(s): ***, ** 1, 5 and 10% significance levels, respectively Source(s): Authors own creation	
	INT IBD NCD WBD BT BT	FDD CEOAGE CEOB FCEO FOWN FOWN	D ROE SIZE AGF	Year Industry	County BT*NCD BT*IBD BT*FBD BT*FBD	Rho Log Likelihood LR test 1(p-value) LR test 2(p-value)	obs 1,19 Note(s): **** **. 1, 5 and 10% s Source(s): Authors own creation	rob

Table 5. Tobit results robustness check

them to use discretion to divert resources at the expense of minority shareholders (Francoeur et al., 2021) and view environmental engagement as a means to extract rent from shareholders' resources and hide poor financial performance (Berrone and Gomez-Meija, 2009), CEO nonduality prevents the opportunistic use of environmental activities and an increase of environmental performance related to agency conflicts of the first type. Principal-agent conflicts are less severe in family firms and the positive effect of CEO non-duality is not significant. Moreover, non-dual CEOs are less prone to respond to external pressure for environmental activities as they are more pressured to achieve short-term results (Walls and Berrone, 2017), but in family firms the relevance of non-financial goals may mitigate the pressure for short-term financial performance, rendering the negative effect of CEO non-duality on environmental performance not significant. Consistent with previous research (De Villiers *et al.*, 2011), our findings indicate that board structures characterized by a large presence of independent directors improves environmental performance but that the effect is significant only for non-family firms, supporting H1. Family firms' independent directors are often closely related to the owning family and aligned with its goals (Chen and Jaggi, 2000). They may behave as delegates of the owning family and exert less effective monitoring compared to non-family firms' independent directors (Cuadrado-Ballesteros et al., 2015). Family firms' independent director decisions tend to be led by the controlling family's needs rather than by the expectations of other stakeholders (Chen and Jaggi, 2000), preventing them from making the difference in terms of environmental performance. Consistently, we find that long directors' tenure significantly lowers family firms' environmental performance as it favors collusion between board members and an owning family whilst the effect is never significant for non-family companies. Board size has a significant positive effect on environmental performance for both types of company due to the positive contribution larger boards can provide in terms of different knowledge, experiences and values (Dalton et al., 1999) that may broaden a company's goals, benefiting environmental interests (Cosma et al., 2021).

As for demographic board diversity, our results indicate that the ratio of female directors on the board exerts a significant positive effect on family firms' environmental performance. These findings confirm the positive effect of female directors on environmental performance (Orazalin and Mahmood, 2021; Islam *et al.*, 2022; Galletta *et al.*, 2022; Muhammad and Migliori, 2022) pointing out that the effect is significant only for family companies. The greater flexibility of family firms compared to non-family companies allows them to better exploit the different skills of women directors (Mubarka and Kammerlander, 2022) to the benefit of environmental performance. This result supports H3, is consistent with, and extends to Europe, the results of the limited stream of literature that has analyzed board gender diversity in family firms in Anglo-Saxon settings (Cordeiro *et al.*, 2020; Nadeem *et al.*, 2020).

Moreover, our results do not support the view that the identification of the owner with the firm, and the quest for reputational assets to ensure long-term sustainability, more effectively affects environmental performance in family than in non-family firms, increasing environmental engagement and performance (Berrone *et al.*, 2010; Agostino and Ruberto, 2021). We find a negative relationship between the name of the owner in a company's name and environmental performance, but the effect is significant only for family firms. Literature suggests that when the family's name is part of the firm's name, family members sense of identification with the firm is higher and SEW is more relevant (Sageder *et al.*, 2018; Calabrò *et al.*, 2022, Saeed *et al.*, 2023). Correlation analyses indicate that having the family name in the name of the company is significantly related to the presence of family members on the board, family CEO and family ownership control as well as a higher average board tenure. Therefore, these findings suggest that a high identification between the family and the business may be related to the dark side of SEW, and "SEW may also serve as a driver of self-serving behavior on the side of the family" to the detriment of other stakeholders' interests (Kellermanns *et al.*, 2012, p. 1179) and of environmental performance.

Size is strongly significant for both types of company, confirming that larger firms should be more sensitive than smaller companies to the control of media and stakeholders (Branco and Rodrigues, 2008), and the effect in terms of environmental performance does not differ between family and non-family firms. Our findings indicate that older family firms are likely to improve environmental performance. The result is consistent with research pointing out that the risk aversion of family firms negatively affects their environmental innovation activities in the early stages, when the uncertainty of such activities is greater. However, in the long-run, concern for the survival of the firm becomes dominant and induces family firms to adopt such innovations (Doluca *et al.*, 2018). Our results also point out that the effect of age is more significant for non-family companies. This is probably because, on the one hand, older firms rely on more stable cash flows and profitability and can afford to engage more in social and environmental activities (Withisuphakorm and Jiraporn, 2016), but, on the other hand, various branches of the family, rather than the founder's family or direct descendants, control older family firms. These branches may even be in conflict with each other and short-term economic interest takes over (Le Breton–Miller and Miller, 2013), slowing the growth of environmental activities compared to non-family companies.

Focusing on family firms, we find that an increase in the percentage of family members on the board lowers environmental performance, supporting H4, but controlling for the interaction with board tenure, we find that the effect is significant for long tenured boards. Unlike Stewardship predictions, this result suggests that when directors sit for a longer time on a board where the family is massively present, they may align their decisions to family-oriented objectives to the detriment of other stakeholders and of environmental performance. Consistently, we find that when female directors sit on low-turnover boards their beneficial effect on environmental performance decreases as long-tenured directors may be left in place because they are major owners (Le Breton Miller *et al.*, 2015) and may support a family-centered decision-making process to the detriment of environmental activities. Therefore, our results support H5 regarding the different moderating effect of board tenure on board diversity in family and non-family firms, and within family firms only for demographic board diversity.

6. Conclusions

6.1 Scientific contribution

This paper studies the effect of structural and demographic board diversity on environmental performance in family firms by focusing on the differences between family and non-family companies as well as within family businesses. In doing so, it addresses the call of Barbera *et al.* (2022) to study the effect of governance characteristics on family firms' non-financial performance and contributes to literature in several ways.

The study contributes to environmental performance research by analyzing, for the first time, the effect of independent directors and CEO non-duality in family firms. It also adds to this stream of literature by providing information on the association between gender diversity and family firms' environmental performance in a European setting, for the first time. In doing so, it contributes to environmental research by pointing out that structural board diversity and demographic board diversity differently affect environmental performance according to a firm's nature, that is, family or non-family owned.

Consistently with prior research focused on U.S. companies (Cordeiro *et al.*, 2020; Nadeem *et al.*, 2020) the present study points out that family firms compared to non-family companies are more effective in exploiting women directors' peculiarities to the benefit of environmental performance. Nevertheless, this paper goes beyond previous literature indicating that the positive effect of board gender diversity decreases as board tenure increases.

Previous research (Graafland, 2020) has analyzed the effect of the presence of family members in management on environmental performance basing on three states: fully non-family-managed firms, companies managed by a mix of family and non-family members, fully

family-managed firms finding that environmental performance is better when a mix of family and non-family members manages the business. This paper adds to family firm research by pointing out that as the ratio of family members on the board increases environmental performance decreases, and that the effect is significantly higher as tenure lengthens.

The present study also contributes to board diversity studies. To the best of our knowledge, it is the first piece of research to underline the different effect of board tenure in family and non-family firms. In doing so, it answers the call for expanding the limited literature on board tenure (Qiang *et al.*, 2023). Actually, our findings suggest that board tenure moderates the effect of demographic board diversity on environmental performance in family companies, but the effect does not hold for non-family firms.

This research contributes to Agency theory by highlighting that, in family firms, board tenure negatively moderates the beneficial effect of board diversity on agency conflicts as it develops family-centered group thinking and lowers monitoring capabilities, to the detriment of environmental performance.

It also contributes to Socioemotional wealth research by providing evidence that some SEW dimensions may also not favor a firm's engagement in environmental activities. Namely, a high sense of identification of the family with the business, as well as a direct influence of the family on the board by the means of a strong presence of its members, may limit environmental performance especially when board member turnover is low.

6.2 Implications for theory and practice

Our findings have implications for firms, regulators and sustainable investors.

Firms should take into account the appointment of a greater number of female directors to the board in order to improve a firm's environmental performance as our findings provide evidence that women on the board can strengthen the firm's environmental agenda and benefit its implementation, especially in family firms. This result has relevant implications for family businesses as it indicates that they may rely on female directors' sensitiveness and experiences in order to enhance environmental performance, but that groupthink tends to substitute diverse thinking as director turnover decreases. More generally, our findings suggest that firms should design boards which, in terms of size, representation of independent directors, leadership structure and specific demographic characteristics of their members, can ensure board monitoring capability, the effectiveness of decision-making processes, as well as a balanced orientation toward financial and environmental performance.

Our results have implications at a national and international level for regulators developing legislative initiatives on climate change mitigation and environmental preservation and quality. Actually, our findings indicate that specific board characteristics can foster companies' environmental commitment and improve their environmental performance whilst others can be a constraint for firms facing current and future environmental challenges. The effect of certain structural board characteristics on environmental performance differs in family and in non-family companies and the strong presence of long-tenured family directors has a negative impact, suggesting that entrenchment may prevent this type of board diversity from improving environmental activities. Therefore, regulators should take into account this evidence when setting measures aimed at promoting board diversity as a means to enhance the effectiveness of environmental legislation.

Moreover, our results have implications for investors who engage in sustainable investments, as they should be aware that board structural and demographic characteristics differently affect the environmental performance of family and nonfamily businesses.

6.3 Limitations and future research

This study has several limitations, which represent opportunities for further studies. This paper analyzes the effect of board diversity on an overall measure of environmental performance; future research could engage in a more grained analysis, studying the impact of board diversity on the different components of environmental performance. Moreover, we focus on board independence and gender diversity, controlling for the effect of board size and tenure. The differences we point out in family and non-family firms suggest that it may be of interest to take into account other forms of board diversity, especially other sources of demographic board diversity such as age, education or nationality. Moreover, within family businesses we use the percentage of family members on the board as a type of demographic board diversity related to family influence but other sources of diversity among family directors may affect environmental performance in family firms. Therefore, future research could analyze the effect of education and generational diversity, within family members on the board, on environmental performance.

We focus on a sample from European countries. This choice ensures that we focus on the relationship between board diversity and environmental performance, limiting the possible noise of differences in the legal regime. Future research could investigate whether the economic and institutional environment as well as country specific cultural aspects influences the results.

Notes

- https://europeanfamilybusinesses.eu/about-european-family-businesses/ accessed on 10 March 2023.
- 2. https://earth.org/fast-fashion-companies/ accessed on 21 July 2023.

References

- Agostino, M. and Ruberto, S. (2021), "Environment-friendly practices: family versus non-family firms", Journal of Cleaner Production, Vol. 329, 129689.
- Anderson, R.C. and Reeb, D.M. (2003), "Founding-family ownership, corporate diversification and firm leverage", *Journal of Law and Economics*, Vol. 46 No. 2, pp. 653-684.
- Arena, C., Bozzolan, S. and Michelon, G. (2015), "Environmental reporting: transparency to stakeholders or stakeholder manipulation? An analysis of disclosure tone and the role of the board of directors", *Corporate Social Responsibility and Environmental Management*, Vol. 22 No. 6, pp. 346-361.
- Barbera, F., Hasso, T. and Schwarz, T.V. (2022), "Family governance and firm performance: exploring the intermediate effects of family functioning and competitive advantage", *Journal of Family Business Management*. doi: 10.1108/JFBM-06-2022-0076.
- Barnea, A. and Rubin, A. (2010), "Corporate social responsibility as a conflict between shareholders", *Journal of Business Ethics*, Vol. 97 No. 1, pp. 71-86.
- Baysinger, B. and Hoskisson, R.E. (1990), "The composition of boards of directors and strategic control: effects on corporate strategy", Academy of Management Review, Vol. 15 No. 1, pp. 72-87.
- Berrone, P. and Gomez-Mejia, L.R. (2009), "Environmental performance and executive compensation: an integrated agency-institutional perspective", *Academy of Management Journal*, Vol. 52 No. 1, pp. 103-126.
- Berrone, P., Cruz, C., Gomez-Mejia, L.R. and Larraza-Kintana, M. (2010), "Socioemotional wealth and corporate responses to institutional pressures: do family-controlled firms pollute less?", *Administrative Science Quarterly*, Vol. 55 No. 1, pp. 82-113.

- Berrone, P., Cruz, C. and Gomez-Mejia, L.R. (2012), "Socioemotional wealth in family firms: theoretical dimensions, assessment approaches, and agenda for future research", *Family Business Review*, Vol. 25 No. 3, pp. 258-279.
- Blanzo-Mazagatos, V., Delgado-García, J.B. and Barrero, J.P. (2022), "Involvement of multiple generations in management and internationalization of family firms in Spain: the moderating effect of SEW dimensions", *Journal of Family Business Management*. doi: 10.1108/JFBM-02-2022-0022.
- Branco, M.C. and Rodrigues, L.L. (2008), "Factors influencing social responsibility disclosure by Portuguese companies", *Journal of Business Ethics*, Vol. 83, pp. 685-701.
- Cabrera-Suárez, M.K. and Martín-Santana, J.D. (2015), "Board composition and performance in Spanish non-listed family firms: the influence of type of directors and CEO duality", BRQ Business Research Quarterly, Vol. 18 No. 4, pp. 213-229.
- Caers, R., Du Bois, C., Jegers, M., De Gieter, S., Schepers, C. and Pepermans, R. (2006), "Principal-agent relationships on the stewardship-agency axis", *Nonprofit Management and Leadership*, Vol. 17 No. 1, pp. 25-47.
- Calabrò, A., Cameran, M., Campa, D. and Pettinicchio, A. (2022), "Financial reporting in family firms: a socioemotional wealth approach toward information quality", *Journal of Small Business Management*, Vol. 60 No. 4, pp. 926-960.
- Chen, C.J. and Jaggi, B. (2000), "Association between independent non-executive directors, family control and financial disclosures in Hong Kong", *Journal of Accounting and Public Policy*, Vol. 19 Nos 4-5, pp. 285-310.
- Chrisman, J.J. (2019), "Stewardship theory: realism, relevance, and family firm governance", Entrepreneurship Theory and Practice, Vol. 43 No. 6, pp. 1051-1066.
- Chrisman, J.J. and Patel, P.C. (2012), "Variations in R&D investments of family and nonfamily firms: behavioral agency and myopic loss aversion perspectives", *Academy of Management Journal*, Vol. 55 No. 4, pp. 976-997.
- Corbetta, G. and Salvato, C.A. (2004), "The board of directors in family firms: one size fits all?", Family Business Review, Vol. 17 No. 2, pp. 119-134.
- Cordeiro, J.J., Profumo, G. and Tutore, I. (2020), "Board gender diversity and corporate environmental performance: the moderating role of family and dual-class majority ownership structures", *Business Strategy and the Environment*, Vol. 29 No. 3, pp. 1127-1144.
- Cosma, S., Schwizer, P., Nobile, L. and Leopizzi, R. (2021), "Environmental attitude in the board. Who are the 'green directors'? Evidences from Italy", *Business Strategy and the Environment*, Vol. 30 No. 7, pp. 3360-3375.
- Croci, E., Doukas, J.A. and Gonenc, H. (2011), "Family control and financing decisions", *European Financial Management*, Vol. 17 No. 5, pp. 860-897.
- Cruz, C., Larraza–Kintana, M., Garcés–Galdeano, L. and Berrone, P. (2014), "Are family firms really more socially responsible?", *Entrepreneurship Theory and Practice*, Vol. 38 No. 6, pp. 1295-1316.
- Cuadrado-Ballesteros, B., Rodríguez-Ariza, L. and García-Sánchez, I.M. (2015), "The role of independent directors at family firms in relation to corporate social responsibility disclosures", *International Business Review*, Vol. 24 No. 5, pp. 890-901.
- Dal Maso, L., Basco, R., Bassetti, T. and Lattanzi, N. (2020), "Family ownership and environmental performance: the mediation effect of human resource practices", *Business Strategy and the Environment*, Vol. 29 No. 3, pp. 1548-1562.
- Dalton, D.R., Daily, C.M., Johnson, J.L. and Ellstrand, A.E. (1999), "Number of directors and financial performance: a meta-analysis", Academy of Management Journal, Vol. 42 No. 6, pp. 674-686.
- Daspit, J.J., Chrisman, J.J., Ashton, T. and Evangelopoulos, N. (2021), "Family firm heterogeneity: a definition, common themes, scholarly progress, and directions forward", *Family Business Review*, Vol. 34 No. 3, pp. 296-322.

- Davis, J., Frankforter, S., Vollrath, D. and Hill, V. (2007), "An empirical test of stewardship theory", *Journal of Business and Leadership: Research, Practice, and Teaching (2005-2012)*, Vol. 3 No. 1, pp. 40-50.
- de Burgos Jiménez, J. and Céspedes Lorente, J.J. (2001), "Environmental performance as an operations objective", *International Journal of Operations and Production Management*, Vol. 21 No. 12, pp. 1553-1572.
- De Massis, A., Frattini, F., Majocchi, A. and Piscitello, L. (2018), "Family firms in the global economy: toward a deeper understanding of internationalization determinants, processes, and outcomes", *Global Strategy Journal*, Vol. 8 No. 1, pp. 3-21.
- De Villiers, C., Naiker, V. and Van Staden, CJ. (2011), "The effect of board characteristics on firm environmental performance", *Journal of Management*, Vol. 37 No. 6, pp. 1636-1663.
- Dekker, J. and Hasso, T. (2016), "Environmental performance focus in private family firms: the role of social embeddedness", *Journal of Business Ethics*, Vol. 136 No. 2, pp. 293-309.
- Doluca, H., Wagner, M. and Block, J. (2018), "Sustainability and environmental behaviour in family firms: a longitudinal analysis of environment-related activities, innovation and performance", *Business Strategy and the Environment*, Vol. 27 No. 1, pp. 152-172.
- Donaldson, L. (1990), "The ethereal hand: organizational economics and management theory", Academy of Management Review, Vol. 15 No. 3, pp. 369-381.
- Donaldson, L. and Davis, J.H. (1991), "Stewardship theory or agency theory: CEO governance and shareholder returns", Australian Journal of Management, Vol. 16 No. 1, pp. 49-64.
- Donaldson, L. and Davis, J.H. (1994), "Boards and company performance-research challenges the conventional wisdom", *Corporate Governance: An International Review*, Vol. 2 No. 3, pp. 151-160.
- Dou, J., Su, E. and Wang, S. (2019), "When does family ownership promote proactive environmental strategy? The role of the firm's long-term orientation", *Journal of Business Ethics*, Vol. 158, pp. 81-95.
- Eddleston, K.A. and Kellermanns, F.W. (2007), "Destructive and productive family relationships: a stewardship theory perspective", *Journal of Business Venturing*, Vol. 22 No. 4, pp. 545-565.
- Ellul, A. (2010), "Control motivations and capital structure decision", Working Paper, Kelley School of Business, Indiana University, Bloomington.
- Elmagrhi, M.H., Ntim, C.G., Elamer, A.A. and Zhang, Q. (2019), "A study of environmental policies and regulations, governance structures, and environmental performance: the role of female directors", *Business Strategy and the Environment*, Vol. 28 No. 1, pp. 206-220.
- Ernst, R.A., Gerken, M., Hack, A. and Hülsbeck, M. (2022), "Family firms as agents of sustainable development: a normative perspective", *Technological Forecasting and Social Change*, Vol. 174, 121135.
- Faccio, M. and Lang, L.H. (2002), "The ultimate ownership of Western European corporations", *Journal of Financial Economics*, Vol. 65 No. 3, pp. 365-395.
- Fernandez-Feijoo, B., Romero, S. and Ruiz-Blanco, S. (2014), "Women on boards: do they affect sustainability reporting?", Corporate Social Responsibility and Environmental Management, Vol. 21 No. 6, pp. 351-364.
- Fischer, H.M. and Pollock, T.G. (2004), "Effects of social capital and power on surviving transformational change: the case of initial public offerings", *Academy of Management Journal*, Vol. 47, pp. 463-481.
- Forbes, D.P. and Milliken, F.J. (1999), "Cognition and corporate governance: understanding boards of directors as strategic decision-making groups", *Academy of Management Review*, Vol. 24 No. 3, pp. 489-505.
- Forés, B., Fernández-Yáñez, J.M., Puig-Denia, A. and Boronat-Navarro, M. (2022), "Unveiling the direct effects of family firm heterogeneity on environmental performance", *Sustainability*, Vol. 14 No. 16, 10442.

- Francoeur, C., Labelle, R., Balti, S. and El Bouzaidi, S. (2019), "To what extent do gender diverse boards enhance corporate social performance?", *Journal of Business Ethics*, Vol. 155, pp. 343-357.
- Francoeur, C., Lakhal, F., Gaaya, S. and Saad, I.B. (2021), "How do powerful CEOs influence corporate environmental performance?", *Economic Modelling*, Vol. 94, pp. 121-129.
- Franks, J., Mayer, C., Volpin, P. and Wagner, H.F. (2012), "The life cycle of family ownership: international evidence", *The Review of Financial Studies*, Vol. 25 No. 6, pp. 1675-1712.
- Galletta, S., Mazzù, S., Naciti, V. and Vermiglio, C. (2022), "Gender diversity and sustainability performance in the banking industry", *Corporate Social Responsibility and Environmental Management*, Vol. 29 No. 1, pp. 161-174.
- Gao, L.S. and Connors, E. (2011), "Corporate environmental performance, disclosure and leverage: an integrated approach", *International Review of Accounting, Banking and Finance*, Vol. 1, pp. 1-30.
- García Martín, C.J. and Herrero, B. (2020), "Do board characteristics affect environmental performance? A study of EU firms", *Corporate Social Responsibility and Environmental Management*, Vol. 27 No. 1, pp. 74-94.
- Gavana, G., Gottardo, P. and Moisello, A.M. (2022), "Related party transactions and earnings management in family firms: the moderating role of board characteristics", *Journal of Family Business Management*. doi: 10.1108/JFBM-07-2022-009.
- Gavana, G., Gottardo, P. and Moisello, A.M. (2023), "Board diversity and corporate social performance in family firms. The moderating effect of the institutional and business environment", *Corporate Social Responsibility and Environmental Management*, Vol. 30 No. 5, pp. 2194-2218.
- Geng, L., Lu, X. and Zhang, C. (2023), "The theoretical lineage and evolutionary logic of research on the environmental behavior of family firms: a literature review", *International Journal of Environmental Research and Public Health*, Vol. 20 No. 6, p. 4768.
- Gómez-Mejía, L.R., Haynes, K.T., Núñez-Nickel, M., Jacobson, K.J. and Moyano-Fuentes, J. (2007), "Socioemotional wealth and business risks in family-controlled firms: evidence from Spanish olive oil mills", *Administrative Science Quarterly*, Vol. 52 No. 1, pp. 106-137.
- Golden, B.R. and Zajac, E.J. (2001), "When will boards influence strategy? Inclination× power= strategic change", *Strategic Management Journal*, Vol. 22 No. 12, pp. 1087-1111.
- Gomez-Mejia, L.R., Lannelongue, G., Muñoz-Bullón, F., Requejo, I. and Sanchez-Bueno, M.J. (2019), "Family firms' concern for the environment: does it pay off to pollute less?", Academy of Management, Vol. 2019 No. 1, 14454, Briarcliff Manor, NY.
- Graafland, J. (2020), "Family business ownership and cleaner production: moderation by company size and family management", *Journal of Cleaner Production*, Vol. 255, 120120.
- Guérin, P. and Suntheim, F. (2021), "Firms' environmental performance and the COVID-19 crisis", *Economics Letters*, Vol. 205, 109956.
- Haniffa, R.M. and Cooke, T.E. (2002), "Culture, corporate governance and disclosure in Malaysian corporations", *Abacus*, Vol. 38 No. 3, pp. 317-349.
- Hewa Wellalage, N. and Locke, S. (2011), "Does CEO duality is really matter? Evidence from an emerging market", *Evidence from an Emerging Market (May 29, 2011), Corporate Ownership* and Control, Vol. 8 No. 4, pp. 1-22.
- Hill, C.W. and Snell, S.A. (1988), "External control, corporate strategy, and firm performance in research-intensive industries", *Strategic Management Journal*, Vol. 9 No. 6, pp. 577-590.
- Huang, Y.C., Yang, M.L. and Wong, Y.J. (2016), "The effect of internal factors and family influence on firms' adoption of green product innovation", *Management Research Review*, Vol. 39 No. 10, pp. 1167-1198.
- Hussain, N., Rigoni, U. and Orij, R.P. (2018), "Corporate governance and sustainability performance: analysis of triple bottom line performance", *Journal of Business Ethics*, Vol. 149, pp. 411-432.

- Huybrechts, J., Voordeckers, W. and Lybaert, N. (2013), "Entrepreneurial risk taking of private family firms: the influence of a nonfamily CEO and the moderating effect of CEO tenure", *Family Business Review*, Vol. 26 No. 2, pp. 161-179.
- Ibrahim, N.A. and Angelidis, J.P. (1995), "The corporate social responsiveness orientation of board members: are there differences between inside and outside directors?", *Journal of Business Ethics*, Vol. 14, pp. 405-410.
- Islam, R., French, E. and Ali, M. (2022), "Evaluating board diversity and its importance in the environmental and social performance of organizations", *Corporate Social Responsibility and Environmental Management*, Vol. 29 No. 5, pp. 1134-1145.
- Jensen, M.C. and Meckling, W.H. (1976), "Theory of the firm: managerial behavior, agency costs and ownership structure", Journal of Financial Economics, Vol. 3, pp. 305-360.
- Katmon, N., Mohamad, Z.Z., Norwani, N.M. and Farooque, O.A. (2019), "Comprehensive board diversity and quality of corporate social responsibility disclosure: evidence from an emerging market", *Journal of Business Ethics*, Vol. 157, pp. 447-481.
- Kellermanns, F.W., Eddleston, K.A. and Zellweger, T.M. (2012), "Article commentary: extending the socioemotional wealth perspective: a look at the dark side", *Entrepreneurship Theory and Practice*, Vol. 36 No. 6, pp. 1175-1182.
- Khojastehpour, M. and Johns, R. (2014), "The effect of environmental CSR issues on corporate/brand reputation and corporate profitability", *European Business Review*, Vol. 26 No. 4, pp. 330-339.
- Kuttner, M., Feldbauer-Durstmüller, B. and Mitter, C. (2021), "Corporate social responsibility in Austrian family firms: socioemotional wealth and stewardship insights from a qualitative approach", *Journal of Family Business Management*, Vol. 11 No. 2, pp. 238-253.
- Le Breton-Miller, I. and Miller, D. (2016), "Family firms and practices of sustainability: a contingency view", *Journal of Family Business Strategy*, Vol. 7 No. 1, pp. 26-33.
- Le Breton-Miller, I., Miller, D. and Bares, F. (2015), "Governance and entrepreneurship in family firms: agency, behavioral agency and resource-based comparisons", *Journal of Family Business Strategy*, Vol. 6 No. 1, pp. 58-62.
- Le Breton–Miller, I. and Miller, D. (2006), "Why do some family businesses out–compete? Governance, long–term orientations, and sustainable capability", *Entrepreneurship Theory and Practice*, Vol. 30 No. 6, pp. 731-746.
- Le Breton–Miller, I. and Miller, D. (2013), "Socioemotional wealth across the family firm life cycle: a commentary on 'Family Business Survival and the Role of Boards", *Entrepreneurship Theory and Practice*, Vol. 37 No. 6, pp. 1391-1397.
- Li, J., Zhao, F., Chen, S., Jiang, W., Liu, T. and Shi, S. (2017), "Gender diversity on boards and firms' environmental policy", *Business Strategy and the Environment*, Vol. 26 No. 3, pp. 306-315.
- Liao, L., Luo, L. and Tang, Q. (2015), "Gender diversity, board independence, environmental committee and greenhouse gas disclosure", *The British Accounting Review*, Vol. 47 No. 4, pp. 409-424.
- López-González, E., Martínez-Ferrero, J. and García-Meca, E. (2019), "Corporate social responsibility in family firms: a contingency approach", *Journal of Cleaner Production*, Vol. 211, pp. 1044-1064.
- Lu, J. and Herremans, I.M. (2019), "Board gender diversity and environmental performance: an industries perspective", *Business Strategy and the Environment*, Vol. 28 No. 7, pp. 1449-1464.
- Luan, C.J. and Tang, M.J. (2007), "Where is independent director efficacy?", Corporate Governance: An International Review, Vol. 15 No. 4, pp. 636-643.
- Mallin, C., Michelon, G. and Raggi, D. (2013), "Monitoring intensity and stakeholders' orientation: how does governance affect social and environmental disclosure?", *Journal of Business Ethics*, Vol. 114, pp. 29-43.
- Martins, A.M. and Pires, C.P. (2023), "Family firms and product recalls: an event study for the US automobile industry", *Journal of Family Business Management*. doi: 10.1108/JFBM-06-2023-0084.

- Mauer, D.C. and Sarkar, S. (2005), "Real options, agency conflicts and optimal capital structure", *Journal of Banking and Finance*, Vol. 29, pp. 1405-1428.
- Miller, D. and Le Breton-Miller, I. (2006), "Family governance and firm performance: agency, stewardship, and capabilities", *Family Business Review*, Vol. 19 No. 1, pp. 73-87.
- Miller, D. and Le Breton–Miller, I. (2014), "Deconstructing socioemotional wealth", *Entrepreneurship Theory and Practice*, Vol. 38 No. 4, pp. 713-772.
- Miroshnychenko, I., De Massis, A., Barontini, R. and Testa, F. (2022), "Family firms and environmental performance: a meta-analytic review", *Family Business Review*, Vol. 35 No. 1, pp. 68-90.
- Mitter, C., Duller, C., Feldbauer-Durstmüller, B. and Kraus, S. (2014), "Internationalization of family firms: the effect of ownership and governance", *Review of Managerial Science*, Vol. 8, pp. 1-28.
- Mubarka, K. and Kammerlander, N.H. (2022), "A closer look at diversity and performance in family firms", Journal of Family Business Management. doi: 10.1108/JFBM-12-2021-0155.
- Muhammad, H. and Migliori, S. (2022), "Effects of board gender diversity and sustainability committees on environmental performance: a quantile regression approach", *Journal of Management and Organization*, pp. 1-26.
- Nadeem, M., Gyapong, E. and Ahmed, A. (2020), "Board gender diversity and environmental, social, and economic value creation: does family ownership matter?", *Business Strategy and the Environment*, Vol. 29 No. 3, pp. 1268-1284.
- Neubaum, D.O., Dibrell, C. and Craig, J.B. (2012), "Balancing natural environmental concerns of internal and external stakeholders in family and non-family businesses", *Journal of Family Business Strategy*, Vol. 3 No. 1, pp. 28-37.
- Nguyen, T.H., Elmagrhi, M.H., Ntim, C.G. and Wu, Y. (2021), "Environmental performance, sustainability, governance and financial performance: evidence from heavily polluting industries in China", *Business Strategy and the Environment*, Vol. 30 No. 5, pp. 2313-2331.
- Orazalin, N. and Mahmood, M. (2021), "Toward sustainable development: board characteristics, country governance quality, and environmental performance", *Business Strategy and the Environment*, Vol. 30 No. 8, pp. 3569-3588.
- Paolone, F., Pozzoli, M., Cucari, N. and Bianco, R. (2023), "Longer board tenure and audit committee tenure. How do they impact environmental performance? A European study", *Corporate Social Responsibility and Environmental Management*, Vol. 30 No. 1, pp. 358-368.
- Patro, S., Zhang, L.Y. and Zhao, R. (2018), "Director tenure and corporate social responsibility: the tradeoff between experience and independence", *Journal of Business Research*, Vol. 93, pp. 51-66.
- Post, C., Rahman, N. and Rubow, E. (2011), "Green governance: boards of directors' composition and environmental corporate social responsibility", *Business and Society*, Vol. 50 No. 1, pp. 189-223.
- Post, C., Rahman, N. and McQuillen, C. (2015), "From board composition to corporate environmental performance through sustainability-themed alliances", *Journal of Business Ethics*, Vol. 130, pp. 423-435.
- Purkayastha, S., Veliyath, R. and George, R. (2022), "Type I and type II agency conflicts in family firms: an empirical investigation", *Journal of Business Research*, Vol. 153, pp. 285-299.
- Qiang, W., Wong, S.S., Koh, K. and Tong, Y.H. (2023), "Does board turnover enhance firm performance? A contingency approach", *Corporate Governance: An International Review*, Vol. 31, pp. 405-424.
- Razzak, M.R. and Jassem, S. (2019), "Socioemotional wealth and performance in private family firms: the mediation effect of family commitment", *Journal of Family Business Management*, Vol. 9 No. 4, pp. 468-496.
- Refinitiv (2022), "Environmental, social and governance (ESG) scores from Refinitiv", available at: https://www.refinitiv.com/content/dam/marketing/en_us/documents/methodology/refinitiv-esgscores-methodology.pdf

- Rhoades, D.L., Rechner, P.L. and Sundaramurthy, C. (2000), "Board composition and financial performance: a meta-analysis of the influence of outside directors", *Journal of Managerial Issues*, Vol. 12 No. 1, pp. 76-91.
- Romano, C.A., Tanewski, G.A. and Smyrnios, K.X. (2001), "Capital structure decision making: a model for family business", *Journal of Business Venturing*, Vol. 16 No. 3, pp. 285-310.
- Rondi, E., De Massis, A. and Kraus, S. (2021), "Servitization through open service innovation in family firms: exploring the ability-willingness paradox", *Journal of Business Research*, Vol. 135, pp. 436-444.
- Rousseau, H.E., Berrone, P. and Gelabert, L. (2019), "Localizing sustainable development goals: nonprofit density and city sustainability", *Academy of Management Discoveries*, Vol. 5 No. 4, pp. 487-513.
- Ruini, L., Ciati, R., Marchelli, L., Rapetti, V., Pratesi, C.A., Redavid, E. and Vannuzzi, E. (2016), "Using an infographic tool to promote healthier and more sustainable food consumption: the double pyramid model by barilla center for food and nutrition", *Agriculture and Agricultural Science Procedia*, Vol. 8, pp. 482-488.
- Saeed, A., Riaz, H., Liedong, T.A. and Rajwani, T. (2023), "Does family matter? Ownership, motives and firms' environmental strategy", *Long Range Planning*, Vol. 56 No. 1, 102216.
- Sageder, M., Mitter, C. and Feldbauer-Durstmüller, B. (2018), "Image and reputation of family firms: a systematic literature review of the state of research", *Review of Managerial Science*, Vol. 12, pp. 335-377.
- Samara, G., Jamali, D., Sierra, V. and Parada, MJ. (2018), "Who are the best performers? The environmental social performance of family firms", *Journal of Family Business Strategy*, Vol. 9 No. 1, pp. 33-43.
- Santos, E., Tavares, V., Tavares, F.O. and Ratten, V. (2022), "How is risk different in family and nonfamily businesses? A comparative statistical analysis during the COVID-19 pandemic", *Journal* of Family Business Management, Vol. 12 No. 4, pp. 1113-1130.
- Schulze, W.S., Lubatkin, M.H., Dino, R.N. and Buchholtz, A.K. (2001), "Agency relationships in family firms: theory and evidence", Organization Science, Vol. 12 No. 2, pp. 99-116.
- Sendlhofer, T. (2020), "Decoupling from moral responsibility for CSR: employees' visionary procrastination at a SME", *Journal of Business Ethics*, Vol. 167 No. 2, pp. 361-378.
- Shakil, M.H., Munim, Z.H., Zamore, S. and Tasnia, M. (2022), "Sustainability and financial performance of transport and logistics firms: does board gender diversity matter?", *Journal of Sustainable Finance and Investment*. doi: 10.1080/20430795.2022.2039998.
- Shaukat, A., Qiu, Y. and Trojanowski, G. (2016), "Board attributes, corporate social responsibility strategy, and corporate environmental and social performance", *Journal of Business Ethics*, Vol. 135, pp. 569-585.
- Shleifer, A. and Vishny, R.W. (1986), "Large shareholders and corporate control", Journal of Political Economy, Vol. 94 No. 3, Part 1, pp. 461-488.
- Sundaramurthy, C. and Lewis, M. (2003), "Control and collaboration: paradoxes of governance", Academy of Management Review, Vol. 28 No. 3, pp. 397-415.
- Tao-Schuchardt, M. and Kammerlander, N. (2023), "Board diversity in family firms across cultures: a contingency analysis on the effects of gender and tenure diversity on firm performance", *Journal of Family Business Strategy*, 100554.
- Vafeas, N. (2003), "Length of board tenure and outside director independence", Journal of Business Finance and Accounting, Vol. 30 Nos 7-8, pp. 1043-1064.
- Veltri, S., Mazzotta, R. and Rubino, F.E. (2021), "Board diversity and corporate social performance: does the family firm status matter?", *Corporate Social Responsibility and Environmental Management*, Vol. 28 No. 6, pp. 1664-1679.
- Villalonga, B. and Amit, R. (2006), "How do family ownership, control and management affect firm value?", *Journal of Financial Economics*, Vol. 80 No. 2, pp. 385-417.

Walls, J.L. and Berr	one, P.	(2017), "The po	wer of one to ma	ke a diffe	erence: how i	nformal	and fo	rmal
CEO power	affect	environmental	sustainability",	Journal	of Business	Ethics,	Vol.	145,
pp. 293-308.								

- Walls, J.L., Berrone, P. and Phan, P.H. (2012), "Corporate governance and environmental performance: is there really a link?", *Strategic Management Journal*, Vol. 33 No. 8, pp. 885-913.
- Westphal, J.D. and Zajac, E.J. (1995), "Who shall govern? CEO/board power, demo- graphic similarity, and new director selection", Administrative Science Quarterly, Vol. 40 No. 1, pp. 60-83.
- Withisuphakorn, P. and Jiraporn, P. (2016), "The effect of firm maturity on corporate social responsibility (CSR): do older firms invest more in CSR?", *Applied Economics Letters*, Vol. 23 No. 4, pp. 298-301.
- Zaman, R., Asiaei, K., Nadeem, M., Malik, I. and Arif, M. (2023), "Board demographic, structural diversity, and eco-innovation: international evidence", *Corporate Governance: An International Review.* doi: 10.1111/corg.12545.
- Zhang, L. (2012), "Board demographic diversity, independence, and corporate social performance", Corporate Governance: The International Journal of Business in Society, Vol. 12 No. 5, pp. 686-700.

Corresponding author

Anna Maria Moisello can be contacted at: annamaria.moisello@unipv.it

For instructions on how to order reprints of this article, please visit our website: www.emeraldgrouppublishing.com/licensing/reprints.htm Or contact us for further details: permissions@emeraldinsight.com